RESEARCH ARTICLE

# A holistic approach to the mycetoma management

Sahar Mubarak Bakhiet<sup>1,2</sup>, Ahmed Hassan Fahal<sup>1</sup>\*, Ahmed Mudawi Musa<sup>1,2</sup>, El Samani Wadaa Mohamed<sup>1</sup>, Rowa Fathelrahman Omer<sup>1</sup>, Eiman Siddig Ahmed<sup>1</sup>, Mustafa El Nour<sup>3</sup>, El Rayah Mohamed Mustafa<sup>3</sup>, Manar El Sheikh A. Rahman<sup>4</sup>, Suliman Hussein Suliman<sup>5</sup>, Mohamed A. Gadir El Mamoun<sup>6</sup>, Hajo Mohamed El Amin<sup>7</sup>

1 The Mycetoma Research Centre, University of Khartoum, Khartoum, Sudan, 2 Department of Molecular Biology, Institute of Endemic Diseases, University of Khartoum, Khartoum, Sudan, 3 Department of Medical Imaging, Soba University Hospital, University of Khartoum, Khartoum, Sudan, 4 Department of Biostatistics, Faculty of Mathematical Sciences, University of Khartoum, Khartoum, Sudan, 5 Department of Surgery, Faculty of Medicine, University of Khartoum, Khartoum, Sudan, 6 The Minister of Health, Ministry of Health, Sennar State, Sennar, Sudan, 7 Department of Surgery, Faculty of Medicine, Sennar University, Sennar, Sudan

\* ahfahal@hotmail.com, ahfahal@uofk.edu

# Abstract

Mycetoma, one of the badly neglected tropical diseases, it is a localised chronic granulomatous inflammatory disease characterised by painless subcutaneous mass and formation of multiple sinuses that produce purulent discharge and grains. If untreated early and appropriately, it usually spread to affect the deep structures and bone resulting in massive damage, deformities and disabilities. It can also spread via the lymphatics and blood leading to distant secondary satellites associated with high morbidity and mortality. To date and despite progress in mycetoma research, a huge knowledge gap remains in mycetoma pathogenesis and epidemiology resulting in the lack of objective and effective control programmes. Currently, the available disease control method is early case detection and proper management. However, the majority of patients present late with immense disease and for many of them, heroic substantial deforming surgical excisions or amputation are the only prevailing treatment options. In this communication, the Mycetoma Research Center (MRC), Sudan shares its experience in implementing a new holistic approach to manage mycetoma patients locally at the village level. The MRC in collaboration with Sennar State Ministry of Health, Sudan had established a region mycetoma centre in one of the endemic mycetoma villages in the state. The patients were treated locally in that centre, the local medical and health personals were trained on early case detection and management, the local community was trained on mycetoma advocacy, and environmental conditions improvement. This comprehensive approach had also addressed the patients' socioeconomic constraints that hinder early presentation and treatment. This approach has also included the active local health authorities, community and civil society participation and contributions to deliver the best management. This holistic approach for mycetoma patients' management proved to be effective for early case detection and management, optimal treatment and treatment outcome and favourable disease prognosis. During the study period, the number of patients



# OPEN ACCESS

**Citation:** Bakhiet SM, Fahal AH, Musa AM, Mohamed ESW, Omer RF, Ahmed ES, et al. (2018) A holistic approach to the mycetoma management. PLoS Negl Trop Dis 12(5): e0006391. <u>https://doi.</u> org/10.1371/journal.pntd.0006391

**Editor:** Carlos Franco-Paredes, Hospital Infantil de Mexico Federico Gomez, UNITED STATES

Received: December 6, 2017

Accepted: March 18, 2018

Published: May 10, 2018

**Copyright:** © 2018 Bakhiet et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

**Funding:** The authors received no specific funding for this work.

**Competing interests:** The authors have declared that no competing interests exist.

with massive lesions and the amputation rate had dropped and that had reduced the disease medical and socioeconomic burdens on patients and families.

# Author summary

Mycetoma enjoys all the neglected tropical diseases (NTDs) characteristics. It frequently affects the poorest of the poor in poor communities in remote regions. The affected population are of low socio-economic status, of low visibility and low political and social voice and hence they are neglected. The disease is considered as a social stigma in particularly among females and children thus they tend to hide it for prolong period and when they are compelled to seek medical care the condition is then at a late stage. The mycetoma patients have many financial constraints that hinder them from seeking medical and health care. In the remote mycetoma endemic areas, the health and medical facilities are meagre, and it is difficult for the patients to reach the regional health centres and thus the majority of patients present with late advanced disease. To overcome these treatment difficulties, the MRC had adopted this holistic management approach to decentralised the patient's care, improve the disease awareness and advocacy, provide free medical and surgical treatment locally at the village level, and to improve the affected villages environmental and hygienic conditions. In this communication, the MRC is reporting on this unique experience, discussing the advantages and difficulties faced it and suggesting recommendations to improve it to be adopted worldwide. Reviewing the medical literature revealed, no report on such management approached for mycetoma patients and thus it worth reporting it.

# Introduction

Mycetoma is a common neglected tropical disease, reported worldwide but endemic in many tropical and subtropical regions in what is known as the mycetoma belt and Sudan seems to have the highest endemicity [1, 2]. It is a chronic granulomatous inflammatory disease caused by several true fungi or certain actinomycetes, and hence it is classified as eumycetoma and actinomycetoma respectively [3, 4]. More than 70 organisms are incriminated in causing mycetoma [5, 6]. It is believed that these causative organisms, which are soil inhabitants, are implanted in the subcutaneous tissue via traumatic inoculation [7, 8]. Mycetoma usually spread to involve the skin, deep structures and bones leading to devastating destruction, deformities and disability [9, 10]. Early localised disease is amenable to cure and good prognosis, however, the late advanced disease is characterised by high morbidity and can be fatal [11, 12, 13]. Mycetoma has serious medical, health and socioeconomic bearings on patients, families, and communities particularly in endemic areas [14, 15, 16].

To date, its global incidence and prevalence are not well documented as mycetoma is a neglected disease, not a notified or a reportable one. Furthermore, in most of the endemic regions, there is no proper disease surveillance system, especially in Sudan. Thus most of the reported cases are limited to anecdotal case reports and passive case detection [17, 18]. More-over, the disease susceptibility, resistance and route of infection are not well characterised [19, 20]. Nevertheless, mycetoma is commonly seen in communities of poor hygiene and environmental conditions where population live in proximity to animals and their dungs. It is believed that thorn pricks and minor injuries are important routes of mycetoma infection. This is

supported by the facts that mycetoma is seen more frequently in the feet of patients of low socioeconomic status, with poor hygiene and in villages with animals enclosures made of thorny trees [21, 22].

Clinically, mycetoma starts as a small painless subcutaneous mass that gradually increases in size, then multiple sinuses with seropurulent discharge that contained grains of different colour and sizes develop [23, 24, 25, 26]. Most patients present late with advanced disease and serious complications due to the painless nature of the disease, patients' low health education level and lack of health facilities in endemic areas [27, 28, 29]. It affects all age groups, but children and young adults of low socio-economic status are affected most, leading to serious economic and social consequences [30, 31, 32].

The proper treatment of mycetoma depends on mycetoma type and disease extent. Numerous mycological and molecular tests are required to identify the causative organisms, and that include grain microscopy and culture, cyto-histopathological examinations and PCR identification [33, 34, 35, 36, 37, 38]. Various imaging techniques such as X-ray, ultrasound, MRI, CT scans are required to determine the disease spread along the various body planes [39, 40, 41, 42]. However, most of these tests and techniques are invasive, of low specificity and sensitivity, expensive for patients and health providers in endemic areas [37]. Currently, there is no point of care diagnostic test for mycetoma. Patients need to travel for long distances to regional centres to establish the diagnosis, and that is not always feasible due to their low socio-economic status, low health education, and roadblocks in particularly during the raining season.

Patients need prolonged periods of management involving diagnosis, treatment both medical and surgical and regular follow up. Treatment may last at least one year for the minor lesions to resolve and several years for large lesions. Even after full recovery, patients need to be followed up closely for evidence of recurrence, which is not uncommon [43, 44, 45, 46].

The currently available treatments for mycetoma are suboptimal and disappointing, characterised by low cure rate (28%) and high patients' follow up dropout (54%) rate [47, 48]. In general, actinomycetoma is treated by a combination of antibiotics, and for eumycetoma, a combination of antifungals and wide local surgical excision is needed. The available medication is not very effective, expensive, with many side effects and hence the high patients' dropout rate [47, 48].

The diagnostic tests and treatment are expensive that amount up to \$2500 per year. Whereas to the annual income in Sudan is less than \$400 per capita (according to the UNDP, 2006), that creates an enormous economic burden on the patients, their families, community and eventually the whole health system in the country [49].

It is interesting to note, worldwide, there are neither preventive or control measures nor programs for mycetoma [20]. The disease surveillance, especially in Sudan, is limited to targeted prevalence studies, case reports, and passive case detection. The absence of a standard-ised, centralised mycetoma surveillance system has far-reaching effects on how the existing interventions are delivered in a cost-effective and evidence-based manner.

In summary, mycetoma is a very devastating endemic disease of the most underprivileged population whether socially, economically or in terms of development. Mycetoma patients tend to travel from distant remote parts of the country to central centres for the treatment. This causes high financial burden and delay in treatment initiation. Furthermore, the disabling nature of the disease hinders access to healthcare service for the majority of the patients. Thus health services decentralisation will improve the accessibility and equity of health services to patients and will directly drop the huge financial burden.

With this background, this community-based study was conducted with the objective of applying a new holistic approach to the management of mycetoma patients at the village level. That included setting up a regional mycetoma centre with a telecommunication network,

offering free of charge both medical and surgical treatment at the centre, training of medical and health staff on early case detection and management, community health education, improvement villages' hygiene all these were based on available health system structure and minimum requirement.

# Materials and methods

This is a community-based cross-sectional study which was conducted at Eastern Sennar Governate, Sennar State, Sudan in the period 2015–2017. The Governate is 400 km south of Sudan capital Khartoum. It has 292 villages with a total population of 219,800 inhabitants. In this study, 19 villages in the Governate were surveyed for early mycetoma patients' detection and management. One village; Wad EL Nimear, the highest endemic village, was studied in depth and compared to another village Wad Onsa which is two kilometres apart with less disease prevalence, Table 1, S1 Map. During the study period, four medical and health mobile missions were organised and conducted by the MRC to the studied villages. The mission team consisted of three consultants surgeons, one physician, two consultants radiologists, four surgical registrars, two radiology registrars, one anaesthesiology registrar, one molecular biologist, one epidemiologist, one environmental consultant, one pharmacist, five medical officers, three surgical scrub nurses, two anaesthetic assistants, four laboratory technologists, two nurses, 10 medical students, one photographer, two fine artists and one musician. The study was implemented in partnership with Federal Ministry of Health, Sennar State Government, the local administrative authority and local community leaders and activists to assure the services sustainability and the study outcome execution.

# Early case detection

Early case detection and management required house to house total coverage survey. That was conducted in 19 villages in the Eastern Sennar Governate. The data were collected by well-trained teams of medical officers, house officers, medical students, health care providers and community activists using a digital pre-designed validated closed-ended questionnaire in smart tablets. Computer Assisted Patients Identifier (CAPI) a computer application which was designed for this study was used. To validate the study questionnaire and the CAPI, a small pilot study was conducted before the data collection in a nearby village.

The CAPI is a computer application predesigned for this study to collect data from the study villages and suspected patients. It was designed by an information and technology expert from the Faculty of Mathematical Sciences, University of Khartoum. It was used in computer tablets or smartphones. It can be used offline and online. CAPI was connected to the MRC,

Table 1.	The Wad	Onsa and	Wad EL	Nimear	villages	characteristics.

Characteristics	Wad Onsa village	Wad EL Nimear village
Houses	Houses reasonably organised, spaced, made of mud or bricks	Not organised, overcrowded, made of mud animals dungs
Animals enclosures	Commonly made of iron, Located outside the houses	Commonly made of thorny trees and bushes, within the houses
Animals dungs	Few, outside the houses	Plenty within the houses
Thorny Trees and bushes	Few in certain parts	Plenty at different parts
Hygiene standard	Reasonable	Poor
Sanitation	Reasonable	Poor
Personal hygiene	Reasonable	Poor

https://doi.org/10.1371/journal.pntd.0006391.t001



Fig 1. Photograph showing a surveyor using a computer tablet with the CAPI programme to collect information on a suspected patient.

https://doi.org/10.1371/journal.pntd.0006391.g001

Data Centre system and the data analysis was performed spontaneously, the results can be displayed on Google maps, Fig 1.

The data collection questionnaire had included the suspected patient's demographic characteristics; name, age, state and village localities, lesion site, the presence of mass, sinuses, grain colour, contact address, lesions photographs, the suspected patient's locality geographic coordinates (latitude, longitude, altitude) and the neighbourhood photographs.

All suspected mycetoma patients from the Governate were referred to Wad Onsa Mycetoma Regional Center (WOMRC). The WOMRC was established in 2015 as a partnership between the Federal Ministry of Health, Ministry of Health, Sennar State, MRC and the local community to manage mycetoma patients locally in their region. The centre consists of small surgical operation complex, two wards, pharmacy, laboratory, ultrasound and out-patient suites and telemedicine facility connecting the WOMRC and the MRC, Fig 2.

At WOMRC centre, the patients were managed by the MRC mobile mission team with continuity of care provided by a surgical team from the regional Sennar Teaching Hospital and the resident doctor at the WOMRC in direct contact with the MRC team via the telemedicine facility.

The diagnosis of mycetoma was established by careful clinical examination and lesion ultrasound examination by mobile ultrasound machine (Paolus–UF-760AG) conducted by the consultant radiologists. All mycetoma suspected patients underwent surgical excisions under general or spinal anesthesia by the consultants surgeons and surgical registrars. The histological examination of the surgical biopsies and grains culture were performed at the MRC in Khartoum as described previously [33, 35]. Some patients with massive lesions were referred to the MRC for further assessment and management. All the investigations and treatment



Fig 2. Photograph showing Wad Onsa regional mycetoma centre at East Sennar Governate. https://doi.org/10.1371/journal.pntd.0006391.g002

were provided free of charge. The Sennar Ministry of Health had provided free meals and transportation for the patients and their families

The confirmed mycetoma patient's information was entered into a predesigned patient management record. This included patient's demographic characteristics, diagnostic tests results, management decisions, treatment received, follow-up and final patient treatment results. This information was regularly checked and updated throughout the patient management and follow-up period.

A system for medicines and consumables procurement, delivery and storage was designed. The medicines included antifungals, antibiotics, analgesics, intravenous fluids and anesthetic medicines as well as the surgical and anesthetic consumables. The medicines were procured from the Central Medical Supply Corporate in Khartoum, shipped and stored at the WOMRC pharmacy at optimum conditions. They were dispensed by the local assistant pharmacist. Patients living in remote areas, of low socio-economic status and unable to attend the outpatient's clinic at WOMRC usually receive their medicines by a community activist who dispensed them use a toktoko (large motorcycle). The patients' information, the medicines doses and quantities were regularly registered. All these information and procedures were documented and regularly reported to the MRC.

# Health care providers training

More than 300 care providers; medical assistants, nurses and public health officers were trained on different aspects of mycetoma, which included the disease causation, presentation, diagnosis and treatment, patients' care, referral indications and system, community health education and disease advocacy. The instructional training methods included presentations,



#### Fig 3. Suspected patients' referral card.

https://doi.org/10.1371/journal.pntd.0006391.g003

group discussions, clinical sessions and ultrasound diagnosis demonstration. Suspected patients' referral card was designed and distributed to the trainee, Fig 3. The improvement in knowledge, attitude and practice (KAP) of the trainees was assessed by pre and post-training tests <u>S1 File</u>. The training sessions were conducted at Singa Town, Sennar State capital, WOMRC and in various East Sennar Governate villages.

Medical students from the local university; Sennar University, as well as the University of Khartoum and other health institutes were training at WOMRC, on early patients' detection, referral and management and the CAPI use was conducted

To gain the Sennar State political involvement and support, the training sessions were addressed by the Sennar State Governor and the Minister of Health, Eastern Sennar locality Governor and community leaders.

# Local community involvement

Several meetings with the local villages' leaders, villagers and community activists were conducted at the Wad Onsa village leader's home, the village's mosque and WOMRC to explain the study objectives and to gain their confidence and support. They were actively involved in the mycetoma advocacy and awareness activities. The local Red Crescent volunteers were trained in mycetoma advocacy and took an important role in improving the local environment and hygiene in the affected villages, Fig 4.

A toktok was donated by the MRC to a community activist at Wad EL Nimear village for transporting patients and their medicines between the different villages and for mycetoma advocacy. In appreciation of excellent, active and energetic involvement in mycetoma advocacy and awareness, three Mycetoma Ambassadors from the Sennar State were selected.

# Health education & advocacy

The study Health Education Team was led by social workers from the Association for Aid and Relief, Japan, Khartoum Office, an active NGO in Sudan with several fine artists, musicians and community volunteers. The health care providers, community leaders and activists, school teachers and medical students from the local university were trained to conduct health education and advocacy sessions. Several health education sessions and activities were carried out



Fig 4. (A) Photograph showing the poor community environment, animals in proximity with the villagers within the houses, animals' dungs and dirt. (B) The villagers removing the thorny animals' enclosures, dirt, animals' dungs and burring them to improve the local environment and hygiene.

https://doi.org/10.1371/journal.pntd.0006391.g004

for early active case detection were conducted. The sessions included small group discussion, school visits sessions, video films watching, and interactive open theatre drama, Fig 5.

"Mesaket Story", a drama film documented a mycetoma patient journey from minor infection which was neglected till limb amputation was produced and was shown to more than 2000 individuals at WOMRC and other villages.



Fig 5. The villagers attending health education session, watching video films on mycetoma awareness.

https://doi.org/10.1371/journal.pntd.0006391.g005



**Fig 6.** (A) The classical animals made of thorny trees and bushes covered with animals' dungs, dirt and rubbish. (B)The new modern animals' cages outside the village. https://doi.org/10.1371/journal.pntd.0006391.g006

#### Community environmental improvement campaigns

Several campaigns to improve Wad El Nimear village environment, sanitation, and hygiene to reduce the mycetoma transmission risk factors such as thorns, sharp objects, animals dungs, were organised by the State Government, official local authorities, community leaders and activists, and Red Crescent volunteers in collaboration with the study team. The thorny trees and bushes, thorny animals enclosure, animals dungs, dirt and rubbish, were removed and burnt, Fig 6.

To improve the village hygiene, reduce the contact with the animals and their excreta and to eradicate the thorny cages, 72 modern animal enclosures were constructed outside Wad EL Nimear village. This project was conducted by a kind donation from an engineering company as its social reasonability activity. These new animals cages were distributed free of charge to the villagers, Fig 6.

In mycetoma, the foot is affected most, and traumatic inoculation of the causative organisms which are present in the soil is believed to be the route of infection. The habit of going barefooted in the villages and the minor trauma are considered the risk factors for mycetoma. To reduce these risk factors, the study team has distributed around 800 new shoes to the school pupils at Wad EL Nimear village to improve the personal hygiene and to reduce the risk of developing mycetoma.

# The social impacts of mycetoma

Forty students from the Department of Social Sciences, at the University of Khartoum, spent two weeks at Wad Onsa village studying the social background of the population in the affected villages in the study area and assessed their KAP to mycetoma and its socioeconomic impacts. They surveyed in depth ten villages in the locality. Opened ended questionnaire and focus group discussions were used to obtain the data.

# **Project sustainability**

A Project Management Board was established headed by the Minister of Health and the senior health officials, Sennar State, the Sennar State Mycetoma control programme officer, local villages' leaders and activists, local health care providers and MRC representative. The Board oversees the project implementation and update, problem sharing, analysis and solving. The Board has regular meetings to review the quarterly reports to provide advice and recommendations for improvement.

# **Ethics statement**

The study ethical clearance was obtained from Soba University Hospital Ethical Committee to conduct the study. Informed consents were obtained from the leaders of the villages, informed written consents were obtained from State Ministry of Health and every suspected and confirmed patients. All medical records were anonymised

# Results

# Patients' management

During the study period, 758 mycetoma suspected patients from the surveyed villages and other villages in Eastern Sennar Governate were seen at WOMRC. All of them had an ultrasound examination of the suspected lesions. Of them, 220 patients had ultrasonic evidence of myce-toma, and they underwent wide local surgical excisions (218 patients), and two patients had amputations. They were 134 males (60.9%) and 86 females (39.1%). Their ages ranged between 2 and 70 years and age group 15–30 years was the most affected one. Most of them were students 68 (30.9%), housewives 46 (20.9%), farmers 35 (15.9%), (Table 2). The geographical distribution was uneven, but Wad El Nimear village had the highest prevalence, (Table 2).

Most of the pateints (72.2%) had short disease duration. Pain at the mycetoma site was not a common symptom in these patients; seen in only 39 patients (17.7%). Local trauma at the mycetoma site was reported in only 38 patients (17.3%). Most of the patients had no sinuses (early lesion) 142 (64.5%), and 72 patients (32.7%) had black grains discharge from their sinuses, (Table 2).

The foot 159 (72.2%) and hand 59(26.8%) were affected the most. Less common sites were the back and gluteal one each, (Table 1). The majority of patients 139 (63.2%) had small lesions less than 5 cm in diameter, 51 patients (23.2%) had lesion between 5–10 cm in diameter, and only two patients (0.9%) had lesions more than 10 cm in diameter, (Table 2).

The lesions ultrasound examination findings were mycetoma in 202 patients (91.8%) and foreign body granuloma in 18 patients (8.2%). The surgical procedures performed ranged from wide local excision 218 (99%) to amputation 2 (1%). All patients had an uneventful post-operative recovery. The operatives findings included mycetoma lesions 192 (87.3%), foreign body granulomas with thorns 18 (8.2%), fibroma 2 (1%) and others soft tissue masses. The diagnosis was confirmed by surgical biopsies histopathological examinations, and that showed evidence of eumycetoma in 189 patients (85.9%), foreign body granuloma 17(7.7%), actinomycetoma 3 (1.4%) and others 11(5%). The latter included no-specific granuloma, neuromas and fibromas, (Table 3).

Most of the patients were followed up at the WOMRC. Thirty-seven patients (16.8%) developed recurrence, due to multifactorial factors which included massive lesion, patients' non-compliance with treatment or other factors. Twenty-five patients (11.4%) were lost to follow-up.

# Patients' information system

Confirmed Mycetoma patients' information was entered into the pre-designed patient's management records. These records included full details of the patient's demographic characteristics, diagnostic tests results, the management offered, follow-up and final patient treatment result. This information was regularly monitored and updated throughout the patient journey.

Characteristics	No.	%
Gander		
Male	134	60.9
Female	86	39.1
Age		
>15 years	51	23.2
15–30 years	106	48.2
31-60 years	61	27.7
>60 years	02	0.9
Occupation		
Student	68	30.9
Housewives	54	20.9
Farmers	35	15.9
Workers	05	2.3
Jobless	07	3.2
Other	26	11.8
Missing data	32	14.5
Employee	01	0.5
Locality		
Wad El Nimear	40	18.2
Al-Ragal	22	10
Wad Alabas	22	10
Al-Amarat	20	9.1
Wad Onsa	15	6.8
Other villages	101	45.9
Duration		
<2 years	63	28.6
2–5 years	96	43.6
6-10	28	12.7
>10	13	5.9
Missing	20	9.1
Local pain		
Yes	39	17.7
No	152	69.1
Sometimes	12	5.5
Missing	17	7.7
Local trauma		
Yes	38	17.3
No	151	68.6
Not sure	13	5.9
Missing	18	8.2
Sinuses presence		
No	142	64.5
Yes	51	23.2
Missing	27	12.3
Grains discharge	2,	12.5
Yes	34	15.5
No	161	73.2
110	101	13.4

#### Table 2. The patients' demographic characteristics.

(Continued)

#### Table 2. (Continued)

Characteristics	No.	%
Missing	25	11.3
Site		
Foot	159	72.2
Hand	59	26.8
Other rare sites	02	01.0
Size in cm		
<5 cm	139	63.2
5–10 cm	51	23.2
>10 cm	02	0.9
Missing	28	12.7

https://doi.org/10.1371/journal.pntd.0006391.t002

All these data were systematically reported to the MRC in quarterly basis through two types of reporting format; hard copy and a digital one, the latter one was transmitted through a telemedicine facility at the WOMRC and MRC. The reported information was systematically entered in the pre-designed data analysis software for further analysis and systematically checked for information accuracy. Data from management teams, diagnostic services and inventory was crosschecked and discussed regularly to improve recording and reporting process.

Findings	No.	%
Ultrasound findings		
Mycetoma	202	91.8
Foreign Body Granuloma	18	08.2
Surgical treatment		
Wide local excision	218	99
Amputation	12	01
Operative findings		
Mycetoma	192	87.3
Foreign body granuloma	18	8.2
Fibroma	02	01
Others	08	3.3
Final diagnosis		
Eumycetoma	189	85.9
Foreign body granuloma	17	7.7
Actinomycetoma	03	1.4
Others	11	05
Post-operative Recurrence		
Yes	37	16.8
No	157	71.4
Miss	26	11.8
Follow up		
Dropout	26	11.8

#### Table 3. The patients' management outcome.

https://doi.org/10.1371/journal.pntd.0006391.t003



#### Free medicines management

The current treatment of choice for eumycetoma is itraconazole in a dose of 400mg /day. It costs around 26 US\$/day, that is not affordable by neither patients nor local health authorities, and hence the MRC managed to raise funds to procure and dispense itraconazole free of charge to patients at the WOMRC.

A system for medicines procurement, delivery, storage and dispensing at the WOMRC was designed and tested during the study.

#### Community mycetoma advocacy activities

A random sample of 218 individuals were tested before and after showing them "Mesaket Story" a drama film. The results showed improvement in their knowledge, attitude and practice and towards mycetoma, (Table 4).

Several small group sessions were organised at different villages, schools, mosques and community clubs. 200 community activists, 50 Red Crescent volunteers and 500 school teachers were trained on mycetoma advocacy and awareness.

The fine artists and musician had organised several interactive open theatre dramas. Different health education materials in different forms were used.

Experts in watercolouring, oil painting and photography have greatly contributed to mycetoma advocacy and awareness through their production of high-quality paintings, photographs and videos captured from the studied Governate, Fig 7.

# The social aspects

During the study, six patients with amputations received limb prosthesis donated by the Agent of Aid and Relief, Japan. This had remarkably improved the life quality.

# Discussion

The MRC records showed that the EL Gazeria, White Nile and Sennar States are the top endemic states in the country [28]. Despite been the third, Sennar State has been chosen for the present study due to the strong commitment of the political leaders, civil societies and communities leaders to support and implement the study and its research outcome. The local communities' leaders were aware of the negative impacts of mycetoma on health and its socioeconomic bearings, and hence their response to the study team requests was swift and extremely positive.

The concept of the village specialised mycetoma centre reported in this communication is a unique one. The WOMRC had delivered integrated medical and social services at the heart of an endemic area. The centre was established as a joint project between the Federal, Sennar State Ministries of Health and the local community, which by itself an exceptional initiative. This study demonstrated numerous positive impacts of the centre on the local communities. It provided local, decentralised mycetoma services in a location with bare minimal health service provision and has improved the local population health education and disease awareness.

The telemedicine which links the MRC in Khartoum with the centre has facilitated the management and follow-up of patients, thus reducing the financial and geographic burdens on the patients and families, and also reduced the patients' follow-up dropout rate. The dropout reported in this study (11.8%) is less than that reported at the MRC (54%). Although the capital cost of the telemedicine setup is high in developing countries, however, in the long run, it is cost-effective for the patients, families and health authorities in mycetoma endemic regions. This is a unique experience that has not previously been reported for mycetoma or other neglected tropical diseases (NTD) and can be replicated for other endemic NTDs.

Questions	Response Before filming	- Response After filming
mycetoma is caused by		
Fungus	51 (23.4%)	19 (7.8%)
Bacteria	29 (13 3%)	11 (5.0%)
Both	61 (28 0%) **	180 (82 6%) ** p
Viral	61 (28.0%) **	08 (3 7%)
Not sure	16 (7 3%)	00(0%)
Mycetoma causative organism found in	10 (7.570)	
Soil	193 (88 5%) **	214 (98 2%)
Water	15 (6 9%)	03 (1 3%)
Other	00 (00%)	00 (00%)
I do not know	10 (4 6%)	01(0.5%)
Mycetoma presentation		
Painless swelling	90 (41 3%)	33 (15 1%)
Sinuses formation	45 (20.6%)	11 (5.0%)
Discharge and grains	61 (28%)	26 (11 9%)
Swelling sinuses grains	22 (10 1%)**	148 (67 9%)
Mycetoma route of entry		110 (07.970)
Through small wound in the skin	65 (29.8%)	13 (6.0%)
Following sharp object or thorns injuries	90 (41 3%)	26 (11.9%)
Poor hygiene defecation	41 (18 8%)	05 (2 3%)
All of the above	22 (10.1%)	174 (79 8%)** p
Is it infectious		1/4(/).0/0) p
ves	96 (44 5%)	31 (14 2%)
No	68 (31 3%)	180 (82 6%)** p
Not sure	54 (24 3%)	07 (3 2%)
Mycetoma is seen in	01(21.070)	07 (3.270)
Foot	159 (72 9%)	35 (16 1%)
Hand	28 (12 8%)	02(0.9%)
Back	01 (0 5%)	01(0.5%)
 Head	03 (1.4%)	03(1.4%)
Abdomen	07(3.2%)	00(0.0%)
All of the above	20(9.2%)**	177(81.2%)
Mycetoma treatment		1//(01.2/0)
Traditional	05 (2.3%)	02(0.9%)
Medical	113 (51.8%)	172 (78.9%)
Both	100(45.8%)	44 (20.2%)
Disease complications	100(10.070)	11 (20.270)
Difficulty in walking	20(09.2%)	04 (1.8%)
Deep Infections	11(5.1%)	03 (1.4%)
Amputation	137 (62.8%)	26 (11 9%)
All of the above	50 (22.9%)	185 (84 9%)
Prevention measurements	50 (22.770)	105 (01.770)
Shoe wearing during working	209 (95 9%)**	218 (100%)
Other	00(00%)	00(00%)
Not ouro	00(070)	00(00%)

 $^{**} p < 0.01$ 

https://doi.org/10.1371/journal.pntd.0006391.t004



Fig 7. Community leaders, villagers and health and medical staff training material.

https://doi.org/10.1371/journal.pntd.0006391.g007

The global lack of disease control or elimination programmes due to the unavailability of basic disease epidemiological characteristics has resulted in early case detection and management as the only available method to reduce the disease incidence and prevalence and its community impacts [28]. It is now evident that WOMRC has tremendous bearings on disease management by offering early case detection facilities and free, decentralised medical, health and advocacy services. That is evidenced the fact that many patients (63.2%) with small lesions and patients (72.2%) with disease duration of less than five years were seen at the centre and only two patients underwent amputation during the study period compared patients seen at the MRC [28]. Furthermore, that is supported by the improvement in disease awareness as evidenced by the KAP study results.

The mycetoma onset and progress are usually slow and painless, affecting patients of low socio-economic and health education levels. Hence these patients are different from patients



with other deadly infectious diseases, e.g. malaria, cholera, leishmaniasis, where patients have no other choice but to report early and follow medical instructions [50, 51]. Moreover, mycetoma patients with early lesions differ from patients with large disabling mycetoma lesions. Early lesions are usually tiny and painless thus not interfering with their normal daily activities. Some patients consider it at this stage as a trivial or even normal event. These patients usually have many other more pressing social and economic problems than these tiny lesions, e.g. the short busy seasonal farming session, raising children in poor conditions and others. Most of the patients consider these early lesions are not a priority, and in fact, they believe that treatment will delay undertaking other urgent duties, and this explains the late presentation with massive lesions [28,31].

It is evident from this study that our holistic management has addressed many issues. The community engagement activities have led to active early case detection which is supported by the high number of patients with early disease seen at the WOMRC reported in this study. Such patients have a high chance of cure and were amenable to treatment with a good outcome [47,52].

The immediate access to free treatment at the village level has reduced patients' delay in starting treatment, eliminated patients' geographical and financial burdens, treatment interruption, and reduced the high follow up dropout rate. Treatment interruption can induce drug resistance. It is therefore vital to ensure sustainability and availability of the free mycetoma treatment services.

The health system in Sudan consists of the three levels; the rural, regional and central levels. The medical assistants, the health care providers, are the backbone for the management of mycetoma patients in Sudan at the rural level. Most of them have poor surgical experience and used to operate on the mycetoma patients under local anesthesia and suboptimal conditions. This practice has led to the high recurrence rate which was documented in many reports [1,2,53]. Recurrent disease is usually associated with wide local disease spread. Hence it is difficult to cure and necessitate repeated surgical excisions, numerous deformities and disabilities [1,2]. At Sennar State, most of the medical assistants were successfully trained on the different aspects of mycetoma care, management and referral.

A well-trained multidisciplinary team on mycetoma care was developed in Eastern Sennar region. It consists of a trained surgeon, surgical theatre attendants, anesthetic assistants, pharmacy assistant, ultrasound technicians, nurses, information technology expert, statistical clerks, community leaders and activists. This is an essential step in providing comprehensive and holistic management for the mycetoma patients.

To date, the definitive route of infection in mycetoma is an enigma. However, it is clear that mycetoma incidence is high in areas of poor environmental conditions, among people with poor personal hygiene and people living in proximity to animals and their dungs and where thorns, dirt and mud prevail. Hence this study aimed to improve the living and hygienic standards of Wad EL Nimear village, one of the highly endemic villages in the locality. The local villagers were encouraged to improve their living conditions. To achieve this goal, many advocacy and awareness campaigns were conducted, and new modern hygienic animals' cages were constructed and donated to them free of charge. In support of these measures, the local Governate authorities issued a law banning the presence of animals inside the village. The kind donation of the animals' cages by the engineering company as a response to the intensive mycetoma awareness and advocacy in Sudan.

In this study, the community leaders and activists were actively involved in conveying messages to their community in their own culture and traditions. This was important to accept these holistic disease management procedures. Likewise, the local villagers have actively engaged in promoting their health and improvement of local environmental conditions that believed to be the main source of transmitting mycetoma.

In conclusion, the holistic and comprehensive management approach implemented in this study has improved the mycetoma patients' quality of care in the studied endemic area. More early disease was detected and treated. The treatment interruption rate was reduced thus increasing the cure rate and decreasing the recurrence and hospitalisation rates. This will eventually lead to decrease in the amputation and disability rates. The results obtained from this study suggest that such a study can be expanded to other endemic areas in the country. The MRC, as a WHO Collaborating centre on Mycetoma, will communicate this experience to the WHO to share it with other mycetoma endemic countries and assist in better management, prevention and control of the disease.

# Supporting information

**S1 Checklist. STROBE checklist.** (DOC)

**S1 File. The KAP Pre and post-training tests.** (PDF)

**S1** Map. Sudan map showing Sennar state and Wad Onsa and Wad El Nimear villages. (TIF)

# **Author Contributions**

**Conceptualization:** Sahar Mubarak Bakhiet, Ahmed Hassan Fahal, Ahmed Mudawi Musa, Manar El Sheikh A. Rahman, Suliman Hussein Suliman.

- **Data curation:** Ahmed Hassan Fahal, Rowa Fathelrahman Omer, Mustafa El Nour, El Rayah Mohamed Mustafa, Manar El Sheikh A. Rahman, Suliman Hussein Suliman, Mohamed A. Gadir El Mamoun, Hajo Mohamed El Amin.
- **Formal analysis:** Sahar Mubarak Bakhiet, Ahmed Hassan Fahal, Ahmed Mudawi Musa, El Samani Wadaa Mohamed, Eiman Siddig Ahmed, El Rayah Mohamed Mustafa.
- Funding acquisition: Ahmed Hassan Fahal, Mohamed A. Gadir El Mamoun.
- **Investigation:** Sahar Mubarak Bakhiet, Ahmed Hassan Fahal, El Samani Wadaa Mohamed, Rowa Fathelrahman Omer, Eiman Siddig Ahmed, Mustafa El Nour, El Rayah Mohamed Mustafa, Suliman Hussein Suliman, Hajo Mohamed El Amin.
- Methodology: Sahar Mubarak Bakhiet, Ahmed Hassan Fahal, Ahmed Mudawi Musa, El Samani Wadaa Mohamed, Rowa Fathelrahman Omer, Eiman Siddig Ahmed, Mustafa El Nour, El Rayah Mohamed Mustafa, Manar El Sheikh A. Rahman, Suliman Hussein Suliman, Hajo Mohamed El Amin.
- **Project administration:** Sahar Mubarak Bakhiet, Ahmed Hassan Fahal, El Samani Wadaa Mohamed, Manar El Sheikh A. Rahman, Mohamed A. Gadir El Mamoun.

Resources: Ahmed Mudawi Musa.

Supervision: El Samani Wadaa Mohamed.

Validation: Sahar Mubarak Bakhiet, Ahmed Mudawi Musa, Rowa Fathelrahman Omer.

Visualization: Ahmed Mudawi Musa.

- Writing original draft: Sahar Mubarak Bakhiet, Ahmed Hassan Fahal, El Samani Wadaa Mohamed, Rowa Fathelrahman Omer, Eiman Siddig Ahmed, Mustafa El Nour, El Rayah Mohamed Mustafa, Mohamed A. Gadir El Mamoun, Hajo Mohamed El Amin.
- Writing review & editing: Sahar Mubarak Bakhiet, Ahmed Hassan Fahal, Ahmed Mudawi Musa.

#### References

- 1. Fahal AH. (2004) Mycetoma thorn on the flesh. Trans R Soc Trop Med Hyg. 98(1):3–11. PMID: 14702833
- 2. Fahal AH, Hassan MA. (1992) Mycetoma. Br J Surg. 79(11): 1138–1141. PMID: 1467883
- 3. Bonifaz A, Tirado-Sánchez A, Calderón L, Saúl A, Araiza J, et al. (2014) Mycetoma: experience of 482 cases in a single center in Mexico. PLoS Negl Trop Dis. 21; 8(8):e3102.
- Ahmed AO, van Leeuwen W, Fahal A, van de Sande W, Verbrugh H, et al. (2004) Mycetoma caused by Madurella mycetomatis: a neglected infectious burden. Lancet Infect Dis. 4(9):566–74. https://doi.org/ 10.1016/S1473-3099(04)01131-4 PMID: 15336224
- Fahal AH, van de Sandy WW. (2012) The Epidemiology of mycetoma. Current Fungal Infection Reports. 6(4): 320–326.
- Fahal AH. (2011) Mycetoma in Richard, Guerrant, Walker, Peter, Tropical Infectious Diseases: Principles, Pathogens and Practice. Third edition, Elsevier Publisher, 83, pp. 565–568.
- Fahal AH. (2013) Mycetoma in Williams, Bulstrode, O'Connell, Bailey and Love's Short Practice of Surgery 26E: 26th Edition, Oxford University Press, pp 84–88.
- Fahal AH. (2006) Mycetoma: Clinico-pathological Monograph, University of Khartoum Press, pp 40– 49.
- Ahmed AOA, van de Sande WW, Fahal A, Bakker-Woudenberg I, Verbrugh H, et al. (2007) Management of mycetoma: major challenge in tropical mycoses with limited international recognition. Curr Opin Infect Dis. 20(2):146–51. https://doi.org/10.1097/QCO.0b013e32803d38fe PMID: 17496572
- van Belkum A, Fahal AH, van de Sande WW. (2012) Mycetoma by Madurella mycetomatis: A completely neglected medico-social dilemma. In Hot Topics in Infection and Immunity in Children IX. Springer Science & Business Media, New York.
- Fahal AH, Suliman SH, Gadir AFA, EL Hag IA, EL Amin FI, et al. (1994) Abdominal wall mycetoma: an unusual presentation. Trans R Soc Trop Med Hyg. 88(1):78–80. PMID: 8154012
- Hassan MA, Fahal AH. (2004) Mycetoma: in Tropical Surgery, Kamil R, Lumbly J, Westminster Publication LTD, pp 30–45.
- 13. Fahal AH, Suliman SH. (1994) Clinical presentation of mycetoma. Sudan Med J. 32: 46–66.
- López-Martínez R, Méndez Tovar LJ, Lavalle P, Welsh O, Saúl A, (1992) Epidemiology of mycetoma in Mexico: study of 2105 cases. Gac Med Mex. 128(4):477–81. PMID: <u>1308000</u>
- López-Martínez R, Méndez-Tovar LJ, Bonifaz A, Arenas R, Mayorga J, et al. (2013) Update on the epidemiology of mycetoma in Mexico. A review of 3933 cases. Gac Med Mex. 149(5):586–92. PMID: 24108347
- Ahmed AOA, Adelmann D, Fahal AH, Verbrugh HA, Van Belkum A, et al. (2002) Environmental occurrence of Madurella mycetomatis, major agent of human eumycetoma in Sudan. J Clin Microbiol. 40(3): 1031–1036. https://doi.org/10.1128/JCM.40.3.1031-1036.2002 PMID: 11880433
- 17. van de Sande WW. (2013) Global burden of human mycetoma: a systematic review and meta-analysis. PLoS Negl Trop Dis 7: e2550. https://doi.org/10.1371/journal.pntd.0002550 PMID: 24244780
- 18. Mahgoub ES, Murray IG. (1973) Mycetoma. London. William Heinemann. Medical Books. pp. 2–25.
- van de Sande W, Fahal A, Ahmed SA, Serrano JA, Bonifaz A, et al. (2017) Closing the mycetoma knowledge gap. Med Mycol. Sep 9. doi: doi: 10.1093/mmy/myx061. [Epub ahead of print] PMID: 28992217
- Hay RJ, Fahal AH. (2015) Mycetoma: an old and still neglected tropical disease. Trans R Soc Trop Med Hyg. 109(3):169–70. https://doi.org/10.1093/trstmh/trv003 PMID: 25667234
- 21. Lynch JB. (1964) Mycetoma in the Sudan. Ann R Coll Surg Engl. 35: 319–340. PMID: 14248320
- 22. Fahal A, Mahgoub ES, EL Hassan AM, Abdel-Rahman ME, Alshambaty Y, et al. (2014) A New Model for Management of Mycetoma in the Sudan. PLoS Negl Trop Dis 8(10): e3271. https://doi.org/10.1371/ journal.pntd.0003271 PMID: 25356640

- 23. Aounallah A, Boussofara L, Ben Saïd Z, Ghariani N, Saidi W, et al. (2013) Analysis of 18 Tunisian cases of mycetoma at the Sousse hospital (1974–2010). Bull Soc Pathol Exot. 106(1): 5–8. https://doi.org/10.1007/s13149-012-0233-1 PMID: 22552715
- Dieng MT, Niang SO, Diop B, Ndiaye B. (2005) Actinomycetomas in Senegal: study of 90 cases. Bull Soc Pathol Exot. 98(1):18–20. PMID: 15915967
- Marc S, Meziane M, Hamada S, Hassam B, Benzekri L. (2011) Clinical and epidemiological features of mycetoma in Morocco. Med Mal Infect. 41(3):163–4. https://doi.org/10.1016/j.medmal.2010.11.006 PMID: 21194861
- Nenoff P, van de Sande WW, Fahal AH, Reinel D, Schöfer H. (2015) Eumycetoma and actinomycetoma—an update on causative agents, epidemiology, pathogenesis, diagnostics and therapy. J Eur Acad Dermatol Venereol. 29(10):1873–83. https://doi.org/10.1111/jdv.13008 PMID: 25726758
- Fahal A, Mahgoub ES, EL Hassan AM, Jacoub AO, Hassan D. (2015) Head and Neck Mycetoma: The Mycetoma Research Centre Experience. PLoS Negl Trop Dis (2015) 9(3): e0003587. <u>https://doi.org/ 10.1371/journal.pntd.0003587</u> PMID: 25768090
- Fahal AH EL Hassan AM, Mahgoub ES, Rahman ME. (2015) Mycetoma in the Sudan: The Mycetoma Research Centre Update. PLoS Negl Trop Dis. 2015 Mar 27; 9(3):e0003679. eCollection 2015 Mar.
- Hala Taha, Ahmed Fahal, van de Sande Wendy WJ. (2015) Mycetoma: Epidemiology, Treatment Challenges and Progress. Mycetoma: epidemiology, treatment challenges and progress. Res Rep Trop Med. 6; 31–36.
- Lacroix C, de Kerviler E, Morel P, Derouin F, Feuilhade de Chavin M. (2005) Madurella mycetomatis mycetoma treated successfully with oral voriconazole. Br J Dermatol. 152: 1067–68. <u>https://doi.org/10.1111/j.1365-2133.2005.06534.x PMID: 15888176</u>
- **31.** Fahal AH, Shaheen S, Jones DH. (2014) The orthopaedic aspects of mycetoma. Bone Joint J 96-B: 420–425. https://doi.org/10.1302/0301-620X.96B3.31421 PMID: 24589802
- Maiti PK, Ray A, Bandyopadhyay S. (2007) Epidemiological aspects of mycetoma from a retrospective study of 264 cases in West Bengal. Trop Med Int Health. 2002; 7: 788–92.
- Fahal AH, el Toum EA, el Hassan AM, Mahgoub ES, Gumaa SA. (1995) The host tissue reaction to Madurella mycetomatis: new classification. J Med Vet Mycol 33: 15–17. PMID: 7650573
- Ahmed AO, Mukhtar MM, Kools-Sijmons M, Fahal AH, de Hoog S, et al. (1999) Development of a species-specific PCR-restriction fragment length polymorphism analysis procedure for identification of Madurella mycetomatis. J Clin Microbiol 37: 3175–3178. PMID: 10488173
- Baveja C. (2010). Medical Mycology In: Arora DR, Arora BB, editors. Textbook of Microbiology for Dental Students. 3rd ed. Delhi, India: Arya Publications; 322–323
- 36. Yousif BM, Fahal AH, Shakir MY. (2010) a new technique for the diagnosis of mycetoma using fixed blocks of aspirated material. Trans R Soc Trop Med Hyg 104 (1): 6–9. https://doi.org/10.1016/j.trstmh. 2009.06.015 PMID: 19700179
- Wendy WJ van de Sande, Ahmed H Fahal, Michael Goodfellow, El Sheikh Maghoub, Oliverio Welsh, et al. (2014) The merits and pitfalls of the currently used diagnostic tools in mycetoma. PLoS Negl Trop Dis July 03, 2014.
- de Hoog GS, Adelmann D, Ahmed AO, van Belkum A. (2004) Phylogeny and typification of Madurella mycetomatis, with a comparison of other agents of eumycetoma. Mycoses. 47(3–4):121–30. <u>https://</u> doi.org/10.1111/j.1439-0507.2004.00964.x PMID: 15078428
- Abd El-Bagi ME, Fahal AH. (2009) Mycetoma revisited. Incidence of various radiographic signs. Saudi Med J 30: 529–533. PMID: 19370281
- 40. El Shamy ME, Fahal AH, Shakir MY, Homeida MM. (2012) New MRI grading system for the diagnosis and management of mycetoma. Trans R Soc Trop Med Hyg 106: 738–742. <u>https://doi.org/10.1016/j.</u> trstmh.2012.08.009 PMID: 22981317
- Fahal AH, Sheikh HE, EL Lider MA, Homeida MA, EL Arabi YE, et al. (1997) Ultrasonic imaging in mycetoma. Br J Surg. 78: 765–766.
- Abd El Bagi ME. (2003) New radiographic classification of bone involvement in pedal mycetoma. Am J Roentgenol. 180:665–668
- Welsh O, Al-Abdely HM, Salinas-Carmona MC, Fahal AH (2014) Mycetoma Medical Therapy. PLoS Negl Trop Dis 8(10): e3218. https://doi.org/10.1371/journal.pntd.0003218 PMID: 25330342
- 44. Welsh O, Vera-Cabrera L, Welsh E, Salinas MC (2012) Actinomycetoma and advances in its treatment. Clin Dermatol 30: 372–381. https://doi.org/10.1016/j.clindermatol.2011.06.027 PMID: 22682184
- Mahgoub ES, Gumaa SA. (1984) Ketoconazole in the treatment of eumycetoma due to madurella mycetomi. Trans R Soc Trop Med Hyg. 78: 376–379. PMID: 6087513



- 46. Fahal AH, Rahman IA, El-Hassan AM, Rahman ME, Zijlstra EE. (2011) The safety and efficacy of itraconazole for the treatment of patients with eumycetoma due to Madurella mycetomatis. Trans R Soc Trop Med Hyg 105: 127–132. https://doi.org/10.1016/j.trstmh.2010.11.008 PMID: 21247608
- 47. Zein HA, Fahal AH, Mahgoub el S, El Hassan TA, Abdel-Rahman ME. (2012) Predictors of cure, amputation and follow-up dropout among patients with mycetoma seen at the Mycetoma Research Centre, University of Khartoum, Sudan. Trans R Soc Trop Med Hyg 106: 639–644. <u>https://doi.org/10.1016/j.trstmh.2012.07.003 PMID: 22854685</u>
- **48.** van de Sande WW, Mahgoub ES, Fahal AH, Goodfellow M, Welsh O, et al. (2014) The Mycetoma Knowledge Gap: Identification of Research Priorities. PLoS Negl Trop Dis. 2014 Mar 27; 8(3):e2667. doi: 10.1371/journal.pntd.0002667. eCollection 2014 Mar PMID: 24675533
- Welsh O, Vera-Cabrera L, Salinas-Carmona MC. (2013) Current treatment for nocardia infections. Expert Opin Pharmacother. 14(17):2387–98. <u>https://doi.org/10.1517/14656566.2013.842553</u> PMID: 24093436
- Fitzgerald E, Mlotha-Mitole R, Ciccone EJ, Tilly AE, Montijo JM, et al. (2018) A pediatric death audit in a large referral hospital in Malawi. BMC Pediatr. Feb 21; 18(1):75. https://doi.org/10.1186/s12887-018-1051-9 PMID: 29466967
- Sifuna P, Otieno L, Ogwang S, Ogutu B, Andagalu B, et al. (2018) Cause-specific mortality in the Kombewa health and demographic surveillance systems site, rural Western Kenya from 2011–2015. Glob Health Action. 11(1):1442959. https://doi.org/10.1080/16549716.2018.1442959 PMID: 29502491
- Wadal A, Elhassan TA, Zein HA, Abdel-Rahman ME, Fahal AH (2016) Predictors of Post-operative Mycetoma Recurrence Using Machine-Learning Algorithms: The Mycetoma Research Center Experience. PLoS Negl Trop Dis 10(10): e0005007. https://doi.org/10.1371/journal.pntd.0005007 PMID: 27798643
- 53. Suleiman SH, Wadaella ES, Fahal AH (2016) The Surgical Treatment of Mycetoma. PLoS Negl Trop Dis 10(6): e0004690. https://doi.org/10.1371/journal.pntd.0004690 PMID: 27336736

© 2018 Bakhiet et al. This is an open access article distributed under the terms of the Creative Commons Attribution License:

http://creativecommons.org/licenses/by/4.0/ (the "License"), which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited. Notwithstanding the ProQuest Terms and Conditions, you may use this content in accordance with the terms of the License.

